Author's response to reviews

Title: Autosomal dominant pseudohypoaldosteronism type 1 with a novel splice site mutation in MR gene

Authors:

Kyoko Kanda (apricot_coco@yahoo.co.jp)
Kandai Nozu (nozu@med.kobe-u.ac.jp)
Naoki Yokoyama (nyoko@med.kobe-u.ac.jp)
Ichiro Morioka (ichim@med.kobe-u.ac.jp)
Akihiro Miwa (miwa214@med.kobe-u.ac.jp)
Yuya Hashimura (hashimurayuya@yahoo.co.jp)
Hiroshi Kaito (hkaitoh@med.kobe-u.ac.jp)
Kazumoto Iijima (iijima@med.kobe-u.ac.jp)
Masafumi Matsuo (matsuo@kobe-u.ac.jp)

Version: 7 Date: 3 November 2009

Author's response to reviews:

Thank you very much much to send me the final version.
There is only one point I want you to correct in this version.
About Fig.2-A, the picture of the electrophoresis was reversed to right side left. I don't know why. But I tried to submit correct version again, however, every time it will be reversed.
Could you please correct the picture? Only the picture reversed and the character "a to e" is correctly typed.
From the left side, 1st line is the marker which has more than 6 bands. 2nd and 3rd line show only one band and 4th to 6th line show 2 bands.
Thank you very much for your kindness.

Regards,
Kandai Nozu